Author's response to reviews

Title: Evaluation of bioactive sphingolipids in 4-HPR-resistant leukemia cells

Authors:

Aintzane Apraiz (aintzane.apraiz@ehu.es)
Jolanta K Idkowiak-Baldys (idkowia@musc.edu)
María Dolores Boyano (lola.boyano@ehu.es)
Gorka Pérez-Yarza (gorka.perezyarza@ehu.es)
Yusuf A Hannun (hannun@musc.edu)
Aintzane Asumendi (aintzane.asumendi@ehu.es)

Version: 4 Date: 2 November 2011

Author's response to reviews: see over
To whom it may concern at the BMC Series Editorial Production Team,

Please find below the point-by-point description of the formatting changes made on our manuscript “Evaluation of bioactive sphingolipids in 4-HPR-resistant leukemia cells” MS: 6073482485758454.

Major revisions

Author list: the list of authors in the manuscript has been written exactly as they were in the submission system (i.e. Jolanta Idkowiak-Baldys has been replaced with Jolanta K Idkowiak-Baldys).

References: references are now cited in numerical order.

References: functioning link has been provided for the Lipidomics Core Facility at the MUSC (page 9).

Minor revisions

Highlighting/tracking: it has been removed from the manuscript as indicated.

Figures: we faced some problems during the previous uploading of Figures 5 and 6. Finally it was Tonilynn Manibo, from the Journal Editorial Office, who succeeded in uploading them. Therefore, WE WOULD BE VERY GRATEFUL IF YOU COULD DEAL WITH THOSE 2 FIGURES. Remaining Figures were already cropped.

Typography: we have included the sentence “Values represent the percentage of cells in each quadrant” in the legend of Figure 6A (page 31). Moreover, we checked the manuscript and found a mistake in the legend of the Additional file 1 (page 32) that has been corrected. It was not 3.6 µM but 0.36 µM of CFSE.